Stability of the *w*Mel *Wolbachia* Infection following Invasion into *Aedes aegypti* Populations



Ary A. Hoffmann¹*, Inaki Iturbe-Ormaetxe², Ashley G. Callahan¹, Ben L. Phillips³, Katrina Billington², Jason K. Axford¹, Brian Montgomery², Andrew P. Turley², Scott L. O'Neill²

1 Bio21 Institute, Department of Genetics, The University of Melbourne, Parkville, Victoria, Australia, 2 School of Biological Sciences, Monash University, Clayton, Victoria, Australia, 3 Department of Zoology, The University of Melbourne, Parkville, Victoria, Australia

Abstract

The *w*Mel infection of *Drosophila melanogaster* was successfully transferred into *Aedes aegypti* mosquitoes where it has the potential to suppress dengue and other arboviruses. The infection was subsequently spread into two natural populations at Yorkeys Knob and Gordonvale near Cairns, Queensland in 2011. Here we report on the stability of the infection following introduction and we characterize factors influencing the ongoing dynamics of the infection in these two populations. While the *Wolbachia* infection always remained high and near fixation in both locations, there was a persistent low frequency of uninfected mosquitoes. These uninfected mosquitoes showed weak spatial structure at both release sites although there was some clustering around two areas in Gordonvale. Infected females from both locations showed perfect maternal transmission consistent with patterns previously established pre-release in laboratory tests. After >2 years under field conditions, the infection continued to show complete cytoplasmic incompatibility across multiple gonotrophic cycles but persistent deleterious fitness effects, suggesting that host effects were stable over time. These results point to the stability of *Wolbachia* infections and their impact on hosts following local invasion, and also highlight the continued persistence of uninfected individuals at a low frequency most likely due to immigration.

Citation: Hoffmann AA, Iturbe-Ormaetxe I, Callahan AG, Phillips BL, Billington K, et al. (2014) Stability of the wMel Wolbachia Infection following Invasion into Aedes aegypti Populations. PLoS Negl Trop Dis 8(9): e3115. doi:10.1371/journal.pntd.0003115

Editor: Jason L. Rasgon, The Pennsylvania State University, United States of America

Received April 9, 2014; Accepted July 14, 2014; Published September 11, 2014

Copyright: © 2014 Hoffmann et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. All data are available from www.pearg.com/ datasets/.

Funding: The research was supported by the National Health and Medical Research Council (1037003), the Foundation for the National Institutes of Health through the Grand Challenges in Global Health Initiative of the Bill and Melinda Gates Foundation (FNIH55), and the Australian Research Council (FL100100066). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: ary@unimelb.edu.au

Introduction

Insect populations can evolve to respond rapidly to infectious agents including endosymbionts. Examples include the evolution of host suppressors that alter sex ratio of endosymbionts that cause male killing in *Hypolimnas bolina* [1] and changes in the lifespan effects induced by *w*MelPop *Wolbachia* infection in *Drosophila melanogaster* attenuated through selection [2]. Yet in other cases the effects of endosymbionts on host characteristics seem stable. For instance male-killing *Wolbachia* in *Drosophila innubila* have been present for at least 15000 years and there has been no selection for resistance [3].

In addition to evolution in the host genome, changes can also occur among *Wolbachia* strains. These changes have been studied in the *w*Ri *Wolbachia* infection of *Drosophila simulans* first detected in the 1980s [4] and subsequently spread throughout the world [5]. During this process, there has been attenuation of deleterious effects of the *Wolbachia* on fecundity [6]. However there has been little change in cytoplasmic incompatibility caused by the infection or infection frequency in natural infected populations where a combination of maternal transmission leakage and cytoplasmic incompatibility dictates the expected frequency of the infection [7]. Moreover, genomic data point to little change in

the *w*MelPop infection across 4 years after transfer into *A. aegypti* mosquitoes [8].

The stability of *Wolbachia* infections is critical if such infections are to be successfully used in insect host and disease control. *Wolbachia* infections have been proposed as a novel intervention for the long term to suppress dengue and other arboviruses in endemic areas [9,10]. This strategy depends on *Wolbachia* infections being maintained stably at high levels within natural populations as well as continuing to exhibit virus interference without producing changes in virulence. Although the evolution of virulence is unpredictable, comparative data suggest that virulence is unlikely to increase [11]. However *Wolbachia* might still not stably persist in populations if there are changes in maternal transmission, cytoplasmic incompatibility and/or fitness effects associated with *Wolbachia* infections.

Recently the *w*Mel *Wolbachia* infection originally described from *Drosophila melanogaster* and transferred into *A. aegypti* mosquitoes was successfully introduced into two areas near Cairns [12]. This infection suppresses the ability of mosquitoes to transmit dengue [13] and other arboviruses [14,15] and is expected to persist without further introductions because it causes cytoplasmic incompatibility (CI), the process whereby eggs of uninfected mosquitoes die when uninfected females mate with infected males

Author Summary

The *w*Mel infection is a *Wolbachia* infection introduced into *Aedes aegypti* mosquito populations; *w*Mel can provide potential suppression of dengue if the infection persists in populations. We show that the infection has persisted at a high frequency since its initial introduction into two natural populations in North Queensland, Australia. The infection, which is inherited through females, appears perfectly transmitted to offspring in the field. It has continued to produce a high level of incompatibility when infected males are mated with uninfected females; these females are likely to enter populations at a low frequency as migrants. These features ensure that the infection will stay at a high frequency in populations, but costs prevent its rapid spread.

[13]. The successful invasion points the way to the use of *Wolbachia* to control dengue in disease endemic areas.

Because of the possibility that changes might interfere with the invasion potential of *Wolbachia* and persistence of the infection, it is important to instigate long term monitoring regimes to detect change. Recent research [16] has indicated that dengue interference has not been affected in the period following introduction of the *Wolbachia*. Here we test for the stability of the infection and changes in host fitness effects in the recently invaded areas after 3 years. We also provide the first estimate of maternal transmission of the infection based on field reared material based on the notion that at least in *Drosophila* patterns in field reared hosts can differ substantially to those obtained with laboratory reared material [17].

Methods

Field maternal transmission data

Adult A. aegypti mosquitoes were collected from Gordonvale (GV) and Yorkeys Knob (YK), Queensland in January and April/May, 2013. The mosquitoes were collected from BG-Sentinel traps set in residential locations within each suburb. The live female mosquitoes were taken back to the laboratory and then placed into individual 200 mL containers containing 50 mL of water and covered with mesh. Wet sandpaper was placed around the sides of the container to provide an oviposition substrate. Eggs were collected daily and left to embryonate for at least two days. Once egg laying was complete, each female was stored in ethanol and frozen at -20 °C.

Eggs from each individual female were hatched separately in 500 ml of reverse osmosis water, yeast and TetraMin Tropical Fish food. Mosquitoes were reared in a controlled temperature laboratory at 26 °C with a 12:12 L:D cycle and 55–85% humidity. The larvae were reared until they reached 3rd-4th instar, then stored in ethanol and frozen at -20 °C.

All adult and F1 larvae were tested for *Wolbachia* infection, using the methods previously described in Lee et al. [18]. Adults and larvae were extracted using a 5% Chelex-based method. This involved extracting the whole organism in 1.7 mL tubes using 3 μ L of Proteinase K, 250 μ L of Chelex solution with two 3 mm glass beads for adults and 150 μ L of Chelex solution with four 2 mm glass beads for larvae. This was followed by grinding in a mixer mill. The tubes were then incubated for one hour at 65 °C and then for 10 minutes at 90 °C.

Polymerase chain reactions (PCR) were carried out using the Roche LightCycler 480 system (384-well format) using the RT/HRM (real-time PCR/high-resolution melt) assay [18]. This assay

involves using three primers sets which include Aedes universal primers $(mRpS6_F/mRpS6_R)$, A. aegypti specific primers $(aRpS6_F/aRpS6_R)$ and Wolbachia specific primers $(w1_F/w1_R)$. The crossing point (Cp) and melting point (Tm) of the PCR products from each primer are used to diagnose infection with Wolbachia. A sample was scored as Wolbachia positive when there is robust amplification of all three primer sets mRpS6, aRpS6 and w1, while an A. aegypti sample that is Wolbachia negative would amplify only mRpS6, aRpS6. Each RT/HRM assay was run using three positive Wolbachia controls and three negative Wolbachia controls.

In total, there were 32 females (all from GV) producing a total of 303 F1 offspring in the January collection and 47 females (45 from GV, 2 from YK) producing a total of 498 F1 offspring in the April/May collection. Binomial confidence intervals for maternal transmission rates were computed following Zar [19].

Fecundity and cytoplasmic incompatibility

To evaluate any changes in fecundity effects of the infection and the expression of cytoplasmic incompatibility, we established infected and uninfected populations from field-collected individuals in April-May 2013 from both YK and GV. Females were collected from BG-S traps by visiting the traps daily. The infected YK population was established from the offspring of 6 field collected females and was maintained in the laboratory at a size of 450 individuals before being used in experiments. The uninfected YK population was initiated from a single inseminated female that produced offspring. Unfortunately compatible uninfected females are rare in the field.

For GV the numbers of field females used to establish the colonies were 34 for the infected colony and 1 for the uninfected colony. We also established an infected strain on a diverse Cairns genetic background. This strain (wC20) was the original release strain backcrossed again for 3 generations to field uninfected males from various parts of Cairns.

To test fecundity of the GV and YK mosquitoes, the colonies were blood fed (en masse) and then females were isolated individually from the cage (only blood fed fully engorged females were collected). These females were three generations removed from the field. Females from all treatments were fed on the same human volunteer to remove host effects (as approved by the University of Melbourne Human Ethics Committee (approval 0723847)). Engorged females were placed into 70 ml specimen cups containing 30 ml of water and sandpaper strips lining the inside of the container. Ten females were set up for each infected and uninfected colony from GV and YK, and 5 females from the wC20 colony were also included. Three days later eggs were collected and for 3 days thereafter. Eggs were collated across days. Four females did not lay or only laid a few eggs and these were excluded.

The fecundity data were analyzed with a two-way ANOVA run in IBM SPSS Statistics 22. Only the YK and GV data were included, with population and infection treated as fixed factors. We also analyzed the number of larvae emerging from the effects in the same way. Data were normally distributed by Kolmogorov-Smirnov tests, but we also analyzed the egg and larval numbers after log transformation and this transformation did not affect the outcome of the analyses.

Cl 2010 versus post-release

For cytoplasmic incompatibility, we set up 1.5 l containers each with 7 females and 7 males which were collected as virgins by separating them individually at the pupal stage. Sex was determined at the adult stage by exposing mosquitoes to 4 $^{\circ}$ C

for 1 min to immobilize them. Eight containers (each with 7 females and 7 males) were set up per cross, and 9 crosses were undertaken (Table 1) based on different combinations of the infected and uninfected populations. The mosquitoes were first blood fed on the same human volunteer 9 days after eclosion and then 3 days later eggs were collected using sandpaper ovistrips which were replaced daily and collected over 4 days. This was followed by a second blood feeding with the same volunteer at 16 days with repeated oviposition and a third blood feed (again with the same volunteer) at 23 days. Eggs were conditioned after each laying period by drying eggs for 3 days before immersion into 500 ml of water with yeast added and ½ tablet of TetraMin fish food. Hatch rate was scored after a week and containers were scored for larvae based on between 127 and 369 eggs per cycle.

Field analysis of spatial distribution

BG traps were established throughout the main release areas and mosquitoes were retrieved weekly between 23 April 2012 and 25 October 2013. At YK, 44 trap locations were used in total across this period, but at any point in time mosquitoes were only retrieved from a maximum of 19 traps placed throughout the site (minimum 3 traps, mean 12). At GV, 45 trap locations were used, and mosquitoes retrieved from a maximum of 40 traps in the site (minimum 7, mean 19). Although some traps remained in the same location throughout the trapping period, several were moved when residents no longer wanted them on premises, or when they failed to catch adults over an extended period.

When traps were moved, they were usually moved to nearby locations, leading to around 40 unique localities at each site, which varied in their spacing and spatial independence. To account for the likely non-independence of trap locality, and to get a sense of the spatial scale at which independence arises between localities, we used a spatially-explicit point-level error structure (described below). Otherwise, our analysis consisted of a standard logistic regression in which we treat our observations (the number of *Wolbachia*-infected mosquitoes at location i and time t, n_{il}) as arising from a binomial distribution:

$$n_{it} \sim \text{Binom}(p_{it}, N_{it})$$

where p_{it} is the probability of an individual being infected, and N_{it} is the number of individuals sampled.

We then use a linear model to describe p_{it} as,

$$logit(p_{it}) = a + bt + s_i$$

where *a* is the intercept (initial *Wolbachia* prevalence), *b* describes the change in prevalence over time, and s_i is the spatially-explicit random effect.

Under this random effect, the covariation between sites scales with the distance between sites. To achieve this scaling, we chose a first-order covariance function, $f(d_{ij}) = e^{-\lambda d_{ij}}$, where d_{ij} is the distance between site *i* and site *j*, and controls the rate of decay with distance. The resulting covariance matrix was then scaled by a coefficient of variation, *v*. Thus the vector of random effects is drawn from a multivariate normal distribution,

$S \sim mvNorm(M,C)$

where **M** is a vector of zeros, and **C** is a variance-covariance matrix with elements (c_{ii}) described by

 $c_{ij} = v e^{-\lambda d_{ij}}$

We fitted this model separately for each population using the JAGS Gibb's sampler [20], and used minimally-informative priors (see Table 2, Figs. S1, S2 in Text S1). We sampled the posterior distribution of our parameters using 30,000 samples from each of three chains following a burn-in of 30,000 samples. Convergence was assessed by examination of trace plots for each of the parameters. For simplicity, these analyses were run on data from both sexes combined because our initial logistic regressions on binomial data (depending on whether individuals were infected or uninfected) did not indicate any interactions between sex and time (P>0.3).

We also examined spatial patterns on data accumulated across time for individual traps. These analyses were run separately in the

Table 1. Crosses performed between infected and uninfected lines.

	Constrantia suda					
	1	2	3			
Incompatible						
<i>w</i> GV (♂) × GV (♀)	0 (284.25)	0 (268.12)	0 (302)			
<i>w</i> GV (♂) × C20 (♀)	0 (321.75)	0 (369.25)	0 (221.62)			
wC20 (°) × GV (ǫ)	0 (259)	0 (267.5)	0 (222.5)			
<i>w</i> YK (♂) × YK (♀)	0 (251.62)	0 (220.87)	0 (236.5)			
wYK (୦*) × C20 (♀)	0 (274)	0 (303.12)	0 (272.62)			
wC20 (♂) × YK (♀)	0 (236.87)	0 (186.62)	0 (127.37)			
wC20 (°) × C20 (Q)	0 (302.12)	0 (322.75)	0 (270)			
Compatible						
GV (^) × wGV (Q)	0.77 (281.87)	0.88 (259.12)	0.73 (206.75)			
YK (°) × wYK (ǫ)	0.61 (162)	0.79 (262.12)	0.66 (166.25)			

The "w" denotes infected lines. Compatible crosses are presented separately. Entries are hatch rates, with the average number of eggs per cross scored given in brackets.

doi:10.1371/journal.pntd.0003115.t001

Т

Table 2. Fitted spatially-explicit point-level models for each site.

Gordonvale					
Parameter	Prior	Posterior Qu	Posterior Quantile		
		2.50%	50%	97.50%	
Initial prevalence, a	N(0, 1000)	3.65	4.20	4.75	
Coefficient of time, b	N(0, 1000)	-0.0031	-0.0018	-0.0006	
Variance due to sampling locality, v	1/Gamma(0.001, 0.001)	0.089	5.686	1416.015	
Decay rate of across-locality covariance, λ	Gamma(0.001, 0.001)	0.0001	2.2621	1471.5936	
Yorkey's Knob					
Parameter	Prior	Posterior Quantile			
		2.50%	50%	97.50%	
Initial prevalence, a	N(0, 1000)	2.21	2.67	3.14	
Coefficient of time, b	N(0, 1000)	-0.0011	0.0002	0.0015	
Variance due to sampling locality, v	1/Gamma(0.001, 0.001)	0.023	2.186	2428.956	
Decay rate of across-locality covariance, λ	Gamma(0.001, 0.001)	0.0074	8.8932	1151.1250	

Minimally informative priors were used in all cases, and the central estimate as well as 95% credible interval (2.5–97.5%) is reported for the posterior of each parameter. doi:10.1371/journal.pntd.0003115.t002

two time periods for GV when infections changed (see below). We used ArcGIS 9.3, R ("ape" and "vegan" libraries) and SAM v. 4.0 (Spatial Analysis in Macroecology) [21] in these analyses. To test for spatial patterns in infection frequencies across pooled data, we initially used the overall value of Moran's I computed in R, and then when there was significant spatial structure we computed Moran's I for different distance classes and used 1000 permutations run in SAM to assess the significance of spatial structure at different scales. The association between infection frequency in the traps and geographic distance was also compared through Mantel tests run in SAM. Spatial differences in infection frequencies between locations were visualized through mapping frequencies against trap location points in ArcGIS.

Results

Field maternal transmission data

For the January collection, there were 303 out of 303 positive detections. For the April/May collection, 498 out of 498 were positive detections. In all cases the negative controls indicated no *Wolbachia* and the positive controls all indicated *Wolbachia*. These results suggest perfect or near perfect maternal transmission in the field, with lower confidence intervals of >98% for the January collection and >99% for the April/May collection.

Fitness – fecundity

The fecundity results (Fig. 1) indicate a reduction in egg numbers laid and larvae produced by the infected females compared to uninfected females from GV and YK. In the ANOVA involving only GV and YK, there was a significant effect of the infection ($F_{(1,31)} = 7.32$, P = 0.011) but no effect of population ($F_{(1,31)} = 2.94$, P = 0.096) or interaction between population or infection ($F_{(1,31)} = 0.85$, P = 0.364). There was a tendency for YK mosquitoes to have lower counts than GV mosquitoes but the results were quite variable. Mosquitoes from the infected line maintained in the laboratory showed similar egg counts to those from the infected GV population but these tended to be higher than for YK, which may reflect the fact that the YK lines were initiated with fewer field-collected mosquitoes. For larval counts, the ANOVA indicated a similar pattern with a significant effect of the infection $(F_{(1,31)} = 12.51, P = 0.001)$ but not population $(F_{(1,31)} = 3.22, P = 0.082)$ or an interaction $(F_{(1,31)} = 1.11, P = 0.301)$. These results suggest that the negative effects of the *w*Mel infection on fecundity and larval production have persisted in time.

Cytoplasmic incompatibility

For the compatible crosses, hatch rates were initially 76 and 61%, which then increased in the second gonotrophic cycle and then decreased into the third cycle (Table 1). For the incompatible crosses, no eggs were observed hatching for the first gonotrophic cycle. For each of cycle 2 and 3 there were 2 hatched eggs out of several thousand eggs scored, which did not develop beyond the first instar stage and were therefore scored as incompatible. These results highlight that incompatibility associated with infected males from the field and susceptibility to CI in uninfected females remained complete.



Figure 1. Egg and larval counts of females tested individually in containers. Error bars are standard errors. doi:10.1371/journal.pntd.0003115.g001

Field analysis of spatial distribution

Yorkeys Knob. We had poor resolution on both the scale of spatial independence (controlled by λ), as well as the random effect of sampling locality (controlled by v; Table 2). This poor resolution is not unexpected given that the data are often at the very bounds of the probability scale (mean prevalence across all sites was 0.96), but make it impossible to be clear about the scale of spatial independence in these data. Nonetheless, our best estimate of lambda suggested a spatial covariance that declines to zero at distances around 500 m (Fig. S3 in Text S1) indicating relatively localized structure.

Despite the poor resolution including the spatial terms means that we have controlled for any spatial non-independence between our sampling localities. When we do so, it is clear that prevalence is apparently static at YK (95% credible interval, -0.0011-0.0015: Table 2, Fig. 2, Fig. S2 in Text S1). To more clearly display these temporal patterns, the number of uninfected and infected mosquitoes caught in the traps was combined across 2 week periods to ensure a minimum sample size of 18 mosquitoes (median = 44, maximum = 116) for comparing infection frequencies when looking for trends (Fig. 3). The infection frequency decreased in mid-2012, but then increased again in late 2013 to near fixation. The lowest frequency was observed in mid-December 2012 when it fell to 80.3% (N = 71, binomial confidence intervals 70.6-88.8%). The long-term average infection frequency was 94.0% (confidence intervals 92.9-95.0% based on total data).

To characterize spatial patterns in infection frequency when pooled across time, we computed frequencies for all traps that had at least 12 individuals (median = 36, maximum = 405) and where traps had collected *A. aegypti* on at least 7 occasions (median = 24, maximum = 106). Across all traps, there was a significant difference in the number of infected versus uninfected mosquitoes collected (G = 122.7, df = 27, P<0.001 by Monte Carlo simulation). One trap at the road entry point of YK collected a large number of uninfected mosquitoes (13 out of 16, frequency = 18.8%, confidence intervals 4.4–45.6 %) (Fig. 4). When this point was excluded, the infection frequency ranged from 84.6% to 100% in the traps.

In the spatial analysis using Moran's I, there was no evidence of spatial structure, regardless of whether the trap near the YK entry point was included or excluded from the analysis (P>0.5 in both cases). A Mantel test comparing infection frequencies across traps with geographic distance also failed to find a significant association when the entry point trap was excluded (r = -0.017, P = 0.915). In agreement with the individual time and location point analysis, uninfected individuals therefore appeared scattered throughout YK, perhaps with the exception of the trap at the entry point to YK.

Gordonvale. The infection frequency was higher at Gordonvale than Yorkeys Knob (Table 2; Fig. S1 in Text S1), but there was a downward trend in prevalence at GV (95% credible interval for the coefficient of time, -0.003--0.0006). As in the case of YK, the best estimate of lambda suggest spatial covariance that declines to zero at distances around 500 m (Fig. S3 in Text S1).

We visualized these changes by combining mosquitoes across the 2-week intervals (Fig. 3), when the minimum number of mosquitoes per sample was 20 (median = 139, maximum = 248). The infection frequencies were mostly > 95% in 2012, but then dropped to 82.6% (N = 172, confidence intervals of 76.6–87.9%) in early 2013 and stayed low in the ensuing period before increasing again to values >90% after January. Considering only the 2012 data, the overall infection frequency was 98.2% (confidence intervals 97.7–98.6%) compared to the post-January frequency of 96.2% (confidence intervals 95.3-97.0%) suggesting a slight decrease in the infection rate in 2013.

For the Moran and Mantel tests, we computed frequencies for all traps that had at least 20 individuals across 7 occasions and analyzed data separately for the 2012 and post January 2013 periods given the changes in frequency mentioned above. For 2012, the median number of mosquitoes per trap was 78 (maximum 520) while for 2013 it was 96 (445). Individual trap frequencies ranged from 91.1% to 100% (Fig. 5).

There was significant spatial structure based on Moran's I for both 2012 (I = 0.196, P<0.001) and 2013 (I = 0.047, P = 0.025). In the Mantel tests, there was a significant association between the infection frequency and geographical distance in both 2012 (r = 0.255, P = 0.010) and 2013 (r = 0.359, P = 0.005). The analysis of Moran's I at individual distances suggested traps <300 m apart had a similar frequency of the infection. Areas with a relatively high frequency of uninfected individuals overall tended to occur in the northern part of GV (Fig. 5).

Discussion

The long-term data collected from these areas 2–3 years after releases were initiated, indicate a stable pattern of *Wolbachia* invasion. Although uninfected mosquitoes persist in the areas, the original invasion which led to a rapid increase in infection frequency over a few weeks has been successfully maintained. The infection also has not spread outside the surrounding area at this time, most likely reflecting the relative isolation of these areas and the unstable equilibrium that is likely to exist to prevent an outward spread of *Wolbachia* [13]. During the initial invasion, a few *Wolbachia* were detected in areas adjacent to the release area but these detections were transient [12] and invasion is not expected to occur into an area of high mosquito density [22].

The infection frequency data from Yorkeys Knob and Gordonvale suggest that there are differences in the processes affecting uninfected mosquitoes in the two areas. In YK, the frequency of uninfecteds tended to be higher overall and without any detectable spatial structure or a high incidence of infected in particular traps. The spatial structure in GV was also very weak although Moran's I, when aggregated across time, suggests a higher frequency of uninfecteds in at least one area of the township. The uninfected mosquitoes may reflect movement into the area, or a loss of infection due to imperfect maternal transmission or perhaps curing of infected larvae during development. How likely are these processes and can they contribute to the different dynamics of uninfecteds seen in the release area?

Gordonvale is more isolated than Yorkeys Knob, being located 9 km from the nearest southern residential area of Cairns, whereas YK is only 1-3 km from the adjacent areas of Holloways Beach and Trinity Park. Therefore greater movement may contribute to the higher incidence of uninfecteds in Yorkeys Knob. Local active or passive movement of mosquitoes is evident from mark-releaserecapture experiments conducted in the Cairns area [23], genetic data [24] and the movement of infected individuals outside the release areas [12]. Once uninfected females enter an area, they may contribute uninfected individuals to a population if they are already mated, otherwise they are likely to mate with infected males and express incompatibility leading to inviable eggs. The level of incompatibility seems complete in A. aegypti infected with wMel [13] so only mated females from the surrounding area are expected to have much impact. On the assumption that uninfecteds in the BGS traps arise only from these females and their offspring, this suggests a migration rate of around 3% for GV and 6% for YK. With an estimated adult population of 7-8000 for



Yorkeys Knob

Figure 2. Changes in frequency of uninfected mosquitoes across time within Yorkeys Knob and Gordonvale release sites. Only frequency estimates from samples > 18 are given. doi:10.1371/journal.pntd.0003115.g002

GV and YK based on data collected during the *Wolbachia* releases in the wet season [25], these results may indicate an invasion of 200–500 female mosquitoes into the adult population from the surrounding area.

Do the spatial and temporal patterns of uninfecteds provide any suggestion of migration or other processes contributing to uninfecteds? We found a high incidence of uninfecteds in the trap at the main entrance point of YK, although this did not extend to other traps. Uninfected mosquitoes were found at this trap at a markedly higher frequency than at other traps and this might reflect local movement. However there was no evidence of a higher incidence of uninfecteds at another point in GV near the highway where this might have been expected. If larval curing had been contributing to the uninfected population due to localized environmental conditions, the incidence of uninfected mosquitoes would have been expected to show a spatial pattern as was the case



Figure 3. Sampled prevalence, and trends in prevalence across time within Yorkey's Knob and Gordonvale release sites. The circles represent the raw data – proportion of infected mosquitoes – and the size of each circle denotes relative sample sizes at each site (ranging from 1–47 mosquitoes per sample at Gordonvale, and 1–29 at Yorkey's Knob). The trend line (solid line) and its 95% credible bounds (dashed lines) are also shown. doi:10.1371/journal.pntd.0003115.g003

at Gordonvale but not at Yorkeys Knob. However local spatial patterns might also arise from entry points of mosquitoes from surrounding areas, such as through passive movement in vehicles.

We found no evidence that maternal transmission was leaky under field conditions. This represents the first estimate of maternal transmission under field conditions which was previously shown to be perfect under laboratory conditions [13]. Thus the situation in mosquitoes contrasts to the wRi infection in D. simulans where maternal transmission was perfect in the laboratory [26] but there was leaky transmission in the field [27]. The perfect maternal transmission and CI mean that the persistence and spread of the infection will only be related to the fitness effect of the infection. Previous estimates of the deleterious fitness effects suggest that these are small for wMel and that fecundity/viability costs are around 24% as estimated from field cages although these can be difficult to detect because of feeder effects on fecundity [13]. Although a direct comparison is difficult, our data for GV are based on one feeder and suggest a deleterious effect of a similar magnitude.

Overall, our data point to similar deleterious fitness effects and CI associated with the infection since the release. These results



Figure 4. Distribution of infected (filled) and uninfected mosquitoes retrieved from BG-S traps after releases were terminated at Yorkeys Knob. doi:10.1371/journal.pntd.0003115.g004

suggest that the unstable point for wMel invasion may not have shifted much over three years, although the changes in infection frequency at GV suggest that the infection may not have reached a stable equilibrium yet. There is also no evidence for a shift in



Figure 5. Distribution of infected (filled) and uninfected (unfilled) mosquitoes retrieved from BG-S traps after releases were terminated at Gordonvale across 2012 and 2013. Red areas indicate the frequency of the infected mosquitoes at each point. doi:10.1371/journal.pntd.0003115.g005

dengue interference effects across this period [16]. These results suggest that the wMel infection will remain high across time once an invasion has been successful, and with ongoing fitness costs there is also unlikely to be rapid invasion into surrounding areas.

Supporting Information

Text S1 Fig. S1: Posterior estimate of the intercept (parameter a) for both sites. Fig. S2: Posterior estimate of the change in prevalence with time (parameter b) for both sites. Fig. S3: Pairwise distances within sites (top panel) and decay of covariance with distance (lower panel). Dotted lines show 95% bounds around the decay function and reveal poor resolution with regard to the spatial scale of covariation operating in this dataset. (PDF)

References

- Hornett EA, Charlat S, Duplouy AMR, Davies N, Roderick GK, et al. (2006) Evolution of male-killer suppression in a natural population. Plos Biology 4: 1643–1648.
- Carrington LB, Leslie J, Weeks AR, Hoffmann AA (2009) The popcorn Wolbachia infection of Drosophila melanogaster. Can selection alter Wolbachia longevity effects? Evolution 63: 2648–2657.
- Jaenike J, Dyer KA (2008) No resistance to male-killing Wolbachia after thousands of years of infection. Journal of Evolutionary Biology 21: 1570–1577.
- Hoffmann AA, Turelli M, Simmons GM (1986) Unidirectional incompatibility between populations of *Drosophila simulans*. Evolution 40: 692–701.
- Kriesner P, Hoffmann AA, Lee SF, Turelli M, Weeks AR (2013) Rapid sequential spread of two Wolbachia variants in Drosophila simulans. Plos Pathogens 9: e1003607.
- Weeks AR, Turelli M, Harcombe WR, Reynolds KT, Hoffmann AA (2007) From parasite to mutualist: Rapid evolution of *Wolbachia* in natural populations of *Drosophila*. Plos Biology 5: 997–1005.
- Carrington LB, Lipkowitz JR, Hoffmann AA, Turelli M (2011) A reexamination of *Wolbachia*-induced cytoplasmic incompatibility in California *Drosophila simulans*. Plos One 6: e22565.
- Woolfit M, Iturbe-Ormaetxe I, Brownlie JC, Walker T, Riegler M, et al. (2013) Genomic evolution of the pathogenic *Wolbachia* strain, *w*MelPop. Genome Biology and Evolution 5: 2189–2204.
- McGraw EA, O'Neill SL (2013) Beyond insecticides: new thinking on an ancient problem. Nature Reviews Microbiology 11: 181–193.
- Sinkins SP (2013) Wolbachia and arbovirus inhibition in mosquitoes. Future Microbiology 8: 1249–1256.
- Bull JJ, Turelli M (2013) Wolbachia versus dengue: Evolutionary forecasts. Evolution, Medicine, and Public Health 2013: 197–207.
- Hoffmann AA, Montgomery BL, Popovici J, Iturbe-Ormaetxe I, Johnson PH, et al. (2011) Successful establishment of *Wolbachia* in *Aedes* populations to suppress dengue transmission. Nature 476: 454–U107.
- Walker T, Johnson PH, Moreira LA, Iturbe-Ormaetxe I, Frentiu FD, et al. (2011) The wMel Wolbachia strain blocks dengue and invades caged Aedes aegypti populations. Nature 476: 450–U101.
- van den Hurk AF, Hall-Mendelin S, Pyke AT, Frentiu FD, McElroy K, et al. (2012) Impact of *Wolbachia* on infection with chikungunya and yellow fever viruses in the mosquito vector *Aedes aegypti*. Plos Neglected Tropical Diseases 6: e1892.

Acknowledgments

We thank Angela Caird, Carly Herbertson, Flora Zigterman, Frederico Muzzi, Martin Durkan, Rodney Bagita and Sarah Flenley for their assistance with field entomology and community engagement, and Adrian Gover, Darren Stanford and Laurel Converse for their assistance with sample processing and data management.

Author Contributions

Conceived and designed the experiments: AAH SLO. Performed the experiments: AGC KB JKA IIO. Analyzed the data: AAH BLP. Contributed reagents/materials/analysis tools: BM APT. Contributed to the writing of the manuscript: AAH SLO AGC JKA BLP.

- Blagrove MSC, Arias-Goeta C, Di Genua C, Failloux AB, Sinkins SP (2013) A Wolbachia wMel Transinfection in Aedes albopictus is not detrimental to host fitness and inhibits chikungunya virus. Plos Neglected Tropical Diseases 7.
- Frentiu FDZ, T. Walker, T. Popovici, J. Pyke, A.T. Van den Hurk, A. McGraw, E.A. O'Neill, S.L. (2014) Limited dengue virus replication in field-collected *Aedes aegypti* mosquitoes infected with *Wolbachia*. Plos Neglected Tropical Diseases in press.
- Hoffmann AA, Turelli M, Harshman LG (1990) Factors affecting the distribution of cytoplasmic incompatibility in *Drosophila simulans*. Genetics 126: 933–948.
- Lee SF, White VL, Weeks AR, Hoffmann AA, Endersby NM (2012) Highthroughput PCR assays to monitor *Wolbachia* infection in the dengue mosquito (*Aedes aegypti*) and *Drosophila simulans*. Applied and Environmental Microbiology 78: 4740–4743.
- Zar JH (2010) Biostatistical Analysis, 5th Edition. Upper Saddle River, NJ: Pearson Prentice-Hall.
- Plummer M (2013) rjags: Bayesian graphical models using MCMC. R package version 3–11.
- Rangel TF, Diniz-Filho JAF, Bini LM (2010) SAM: a comprehensive application for Spatial Analysis in Macroecology. Ecography 33: 46–50.
- Barton NH, Turelli M (2011) Spatial waves of advance with bistable dynamics: cytoplasmic and genetic analogues of Allee effects. American Naturalist 178: E48–E75.
- Russell RC, Webb CE, Williams CR, Ritchie SA (2005) Mark-release-recapture study to measure dispersal of the mosquito *Aedes aegypti* in Cairns, Queensland, Australia. Medical and Veterinary Entomology 19: 451–457.
- Endersby NM, Hoffmann AA, White VL, Ritchie SA, Johnson PH, et al. (2011) Changes in the genetic structure of *Aedes aegypti* (Diptera: Culicidae) populations in Queensland, Australia, across two seasons: Implications for potential mosquito releases. Journal of Medical Entomology 48: 999–1007.
- Ritchie SA, Montgomery BL, Hoffmann AA (2013) Novel estimates of Aedes aegypti (Diptera: Culicidae) population size and adult survival based on Wolbachia releases. Journal of Medical Entomology 50: 624–631.
- Hoffmann AA, Turelli M (1988) Unidirectional incompatibility in *Drosophila* simulans - inheritance, geographic variation and fitness effects. Genetics 119: 435–444.
- Turelli M, Hoffmann AA (1995) Cytoplasmic incompatibility in *Drosophila* simulans - dynamics and parameter estimates from natural populations. Genetics 140: 1319–1338.